Author's response to reviews

Title: Patient and health professional views on rehabilitation practices and outcomes following total hip and knee arthroplasty for osteoarthritis: a focus group study

Authors:

Marie D Westby (marie.westby@vch.ca)
Catherine L Backman (catherine.backman@ubc.ca)

Version: 4 Date: 27 April 2010

Author's response to reviews:

Thank you for the opportunity to make these final changes to our manuscript and further improve upon its content and clarity. I have noted below the revisions requested in your last e-mail:

Major revisions

Appendix: Your appendix includes a table. Please either include this as an additional file, and cite it as such in the manuscript text, or remove the lines and include the information as normal text. Appendices should be positioned between the Authors' contributions and Acknowledgements section.

Response: The appendix has been reformatted and correctly positioned within the manuscript.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Header, footer: Please remove these - DONE

Abbreviations: Please include these as a paragraph, not a list - DONE

Tables- please remove the visible vertical lines from your tables - DONE

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission.

As per your offer above to make some minor corrections, I have left the figure as is for you to crop as closely as necessary - I have no expertise in this and would prefer you to make this change.

Tables: Please do not use ranges when citing tables, each table should be cited individually i.e. Tables 3, 4, 5 and 6 - DONE
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance - DONE

Sincerely,

Marie Westby, PT, PhD